

Name of Journal: *World Journal of Clinical Cases*
Manuscript Type: INVITED EDITORIAL

Diagnostic challenges from conflicting results of tests and imaging

Corresponding author: Run Yu, MD, PhD, Division of Endocrinology, UCLA David Geffen School of Medicine, Los Angeles, CA 90095, USA.
runyu@mednet.ucla.edu

5/11/2024

Dear Editor,

Thank you for the review.

With all due respect, I believe the reviewer's suggestions are out of the scope of this editorial. The purpose of this editorial is to comment on a specific article, not to provide detailed information on Li-Fraumeni syndrome or work-up of cardiac mass. The editorial is to provide a larger view on how to reconcile conflicting evidence in diagnosis.

I change the title of the editorial per editor requirement.

Best regards,

Run Yu, MD, PhD
Professor
UCLA David Gaffen School of Medicine.